Author's response to reviews

Title: Self-assessment of intercultural communication skills: a survey of physicians and medical students in Geneva, Switzerland

Authors:

   Patricia Hudelson (patricia.hudelson@hcuge.ch)
   Noelle Junod Perron (noelle.junod@hcuge.ch)
   Thomas Perneger (thomas.perneger@hcuge.ch)

Version: 6 Date: 29 August 2011

Author's response to reviews:

On title page, replaced "Department of Community Medicine and Primary Care with Department of Community Medicine, Primary Care and Emergency Medicine

Page 6: put period after reference number 11

Removed a " (but can't remember what page it was on)

In the 7th line of the discussion, reworded the phrase to read "least competent at intercultural communication skills"